Author's response to reviews

Title: Establishment of serum YKL-40 and SCCA combination for the diagnosis of patients with esophageal squamous cell carcinoma

Authors:

Xin Zheng (zhengxin@sysucc.org.cn)
Shan Xing (xingshan@sysucc.org.cn)
Xiao-Min Liu (liuxm@sysucc.org.cn)
Wen Liu (liuwen@sysucc.org.cn)
Dan Liu (liudan@sysucc.org.cn)
Pei-Dong Chi (chipd@sysucc.org.cn)
Hao Chen (chenhao@sysucc.org.cn)
Shu-Qin Dai (daishq@sysucc.org.cn)
Mu-Sheng Zeng (zengmsh@sysucc.org.cn)
Wan-Li Liu (liuwl@sysucc.org.cn)

Version: 2
Date: 19 February 2014

Author's response to reviews: see over
The authors declare that we have no competing interests. There are no non-financial competing interests (political, personal, religious, ideological, academic, intellectual, commercial or any other) to declare in relation to this manuscript.

In these studies, Xin Zheng and Shan Xing carried out the main work and contributed equally. They participated in the design of the study and performed the statistical analysis and drafted the manuscript. Xiao-Min Liu carried out the immunoassays. Wan-Li Liu and other authors conceived of the study, and participated in its design and coordination and helped to draft the manuscript. All authors read and approved the final manuscript.

We also thank ROCHE and ABBOTT who provided technical support such as instrument and reagent on this study.